Author's response to reviews

Title: Identification of the risk for liver fibrosis on CHB patients using an artificial neural network based on routine and serum markers

Authors:

Danan Wang (wdnyan@163.com)
Qinghui Wang (wqh7632@yahoo.com.cn)
Fengping Shan (profshan9@163.com)
Beixing Liu (lbeixing@yahoo.com)
Changlong Lu (cllu@mail.cmu.edu.cn)

Version: 4 Date: 13 August 2010

Author's response to reviews: see over
MS:1380188429335485

Identification of the risk for liver fibrosis on CHB patients using an artificial neural network based on routine and serum markers
Danan Wang, Qinghui Wang, Fengping Shan, Beixing Liu and Changlong Lu

Dear Editor-in-Chief,

Thank you very much for your suggestion.
We have added the information on the informed consent in Methods-Patients. All the changes made in the revised manuscript are marked in red color.

Should you need further information about this submission, please contact us at your early convenience.

Yours sincerely,

Dan-nan Wang,
Institute of Immunology, China Medical University,
92 North 2nd, Shenyang, 110001, P. R. China.